Author's response to reviews

Title: A simple method for assigning genomic grade to individual breast tumours

Authors:

Kristian Wennmalm (kriwen@ki.se)
Jonas Bergh (Jonas.Bergh@ki.se)

Version: 3 Date: 28 June 2011

Author's response to reviews: see over
To the editorial board

Regarding the manuscript MS: 4861143852230523

We are very happy that you have decided to accept our manuscript. We have included the information requested on the title page. Professor Jonas Bergh is affiliated with the Karolinska Institutet and University Hospital (Stockholm, Sweden) as well as the Medical Oncology Breast Unit (Manchester, UK), but mail should be sent to the former institution, why we have included that address only. Building rather than street names are used within the Karolinska Institutet and University Hospital; no street names appear in the affiliations. Sorry for a somewhat delayed response, it was not possible for us get this done during the previous week; we hope this has not caused problems.

Yours sincerely,

Kristian Wennmalm M.D., Ph.D.
Department of Oncology and Pathology
Karolinska Institute and University Hospital
S-171 76 Stockholm, Sweden